Author's response to reviews

Title: Risk factors for delirium in acutely admitted elderly patients: a prospective cohort study

Authors:

Johanna C Korevaar (j.korevaar@amc.uva.nl)
Barbara C van Munster (b.c.vanmunster@amc.uva.nl)
Sophia E de Rooij (s.e.derooij@amc.uva.nl)

Version: 3 Date: 11 April 2005

Author's response to reviews: see over
The Editor

Amsterdam, April 11, 2005

MS: 1555927045716437 – Risk factors for delirium in acutely admitted elderly patients: a prospective cohort study

Thank you for accepting our manuscript “1555927045716437 – Risk factors for delirium in acutely admitted elderly patients: a prospective cohort study”. Enclosed you will find our point-by-point description of the changes made.

Also on behalf of the other authors,
Yours sincerely,

J.C. Korevaar

Correspondence to:
JC Korevaar,
Dep. Clinical Epidemiology & Biostatistics, Room J1B-209
Academic Medical Centre, UvA,
P.O. Box 22660
1100 DD Amsterdam, The Netherlands
tel: + 31 20 5664489  fax: + 31 20 6912683
e-mail: j.korevaar@amc.uva.nl
Formatting changes requested:

1. We have removed the bold highlighting
2. We have changed conclusion → conclusions (abstract)
3. We have placed the references before the full stop.
4. We have changed conclusion → conclusions (text)
5. We have added all authors in the reference list, so the term et al/is no longer used.

Responses to the reviewers reports:

Reviewer 1:
He had no suggestions or remarks

Reviewer 2:
We have checked the English (by means of a native English speaker).
No other remarks were given by this reviewer.

Reviewer 3:
As suggested by the reviewer, we have deleted the “a” before delirium in the text in case this was redundant.
No other remarks were given by this reviewer.